Light Fidelity for Internet of Things: A Survey

Antonio Petrosino^{a,b,*}, Domenico Striccoli^{a,b}, Oleksandr Romanov^c, Gennaro Boggia^{a,b}, Luigi Alfredo Grieco^{a,b}

^aDepartment of Electrical and Information Engineering, Politecnico di Bari, Bari, Italy ^bCNIT, Consorzio Nazionale Interuniversitario per le Telecomunicazioni

^cKPI, Institute of Telecommunication Systems of the National Technical University of Ukraine "Igor Sikorsky Kyiv Polytechnic Institute", Ukraine

Abstract

Light-Fidelity (LiFi) is quickly emerging as the next-generation communication technology thanks to its unique benefits, such as available spectrum, high data rates, low implementation costs, and inherent beamforming capabilities. As a consequence, it is endorsed by the scientific literature as an appealing innovation for disclosing disruptive services. The wavefront of LiFi technology is very wide: in this manuscript, we focus our attention on the interplay with the Internet of Things. Essentially, LiFi can assist the IoT in interconnecting a massive number of heterogeneous devices by addressing the current Radio Frequency spectrum bottleneck. Moreover, by investigating LiFi and IoT individually, several surveys and review papers testify to the noteworthiness of both technologies. However, to the best of the authors' knowledge, a comprehensive investigation of contributions where both of them interplay is missing. To fill this gap, this survey provides a thorough investigation of all the research areas in which LiFi key features might enhance the upcoming IoT networks. The evaluation of existing literature on LiFi adopted in the IoT domain can be valuable in identifying missing gaps arising from the interaction of these two technologies, as well as proficiently pinpointing future research directions.

Keywords: LiFi, IoT, Survey

1. Introduction

Light Fidelity (LiFi) is a promising Optical Wireless Communication (OWC) technology which relies on Visible Light Communication (VLC) for wireless data transmission and has the potential to address the incumbent Radio Frequency (RF) spectrum scarcity. It has been introduced in [1] as a nm-wave communication system that uses VLC to enable a wireless networking system that offers bidirectional multi user and pointto-multi point communication. Furthermore, it is considered as the consequence of the continuous expansion of the electromagnetic spectrum employment to higher frequencies [2].

LiFi solutions provide several benefits over RF networks. In particular, they may ensure high-speed internet broadband services using Light-Emitting Diodes (LEDs) up to 10 Gbps [3, 4]. Additionally, the unlicensed bandwidth for LiFi transmission is several orders of magnitude greater than the whole RF spectrum. Indeed, visible light includes frequencies ranging from around 400 to 800 THz. As a consequence, its spectrum is 10,000 times broader than the total RF spectrum [5–8]. Furthermore, VLC transmits data utilizing LEDs and a photodiode at the receiving terminal. As a result, it could be applied efficiently in environments where RF transmission may be dangerous, such as aircraft cabins.

The fluctuation in light intensity directly pertains to the information contained in the transmitted message. In LiFi, there are several types of modulation, including single carrier, multiple carrier, and color modulation [9]. For example, the blinking of the LED (even at negligible frequencies for the human eye) can be used for the data transmission. In particular, if the LED is switched on it can be considered as a digital "1". On the contrary, if the LED is switched off, it is treated as a digital "0" [10]. Despite LiFi networks can be easily implemented as a bidirectional half-duplex system in VLC, some intriguing works make use of VLC for the DownLink (DL) and the infrared spectrum for the UpLink (UL) [11]. This is useful to avoid distracting human mobile users without affecting the lighting conditions of the room [12]. Moreover, it prevents interference between the UL and DL, by allowing simultaneous signal broadcasting [13, 14].

Since LEDs are already widely used in homes, factories, and streetlights, LiFi may take use of the existing lighting infrastructure by reducing the deployment fee. This holds true especially for massive Internet of Things (IoT) devices deployments, due to the capability of light-based communication to provide very large bandwidth and support high nodes density, which are important requirements for IoT environments [15]. Moreover, the adoption of LED lamps in LiFi can increase energy efficiency. In fact, as known, LED lamps are characterized by very long lifetimes and reduced energy consumption [16]. So, they can be effectively employed in LiFi systems by integrating data transfer and low-power illumination in a constrained scenario, such as IoT [15, 17, 18]. Even though it is clear that the amount of energy consumed depends on the various system components, the intrinsic low-power nature of

^{*}Corresponding author. *Email address:* antonio.petrosino@poliba.it (Antonio Petrosino)

Preprint submitted to Optical Switching and Networking

LEDs, combined with additional techniques aiming to save energy (that will be discussed more in detail in Section 8), can greatly help to reduce the overall energy consumption in LiFi systems [16, 19–21].

The well-known IoT paradigm defines a network of smart interconnected objects that can sense real-world phenomena and transmit information over the Internet without the need for human interaction [22]. According to [23], in 2022 the IoT market reached around 15 billion active connections. In this context, wireless access is currently mostly accomplished with the aid of RF. However, radio technologies in access networks are already approaching their limit in terms of throughput and Quality of Service (QoS). Specifically, owing to the currently saturated RF spectrum, it is challenging to accommodate the evergrowing amount of devices to be handled in a IoT environment [24]. Furthermore, the application of the RF presents some relevant issues, including poor privacy due to radio waves freely penetrating walls, a huge amount of energy required for signal relaying and service equipment cooling, and limitations in its adoption in some environments like aircraft, hospitals, and specific factories. To cope with these issues, the implementation of LiFi may disclose disruptive services in the IoT domain.

Nevertheless, despite LiFi is still in its early stages and is not yet commonly accessible, numerous teams and researchers are working to create and deploy LiFi-based networking solutions [25]. Consequently, several review and survey papers is discussing the main functionalities of LiFi, but separately from the IoT environment. According to this, and to the best of the authors' knowledge, a thorough analysis of the state-of-the-art in LiFi technology integrated into IoT environments is still missing. To fill this gap, the primary goal of this work is to comprehensively examine all areas in which academics are focusing their efforts on the suitability and potential of the LiFi applied to a highly dynamic environment such as the IoT. Then, based on this analysis, the main challenges and open issues are illustrated, to provide guidelines for future research directions on this theme.

The remainder of the paper is organized as follows. Section 2 explains the rationale at the basis of the classification proposed in this survey. Section 3 illustrates the main topics covered by other similar surveys and review papers. Section 4 highlights the main differences and missing gaps filled by this work if compared to them. Sections 5-10 provide a detailed description of the contributions provided by all the surveyed papers, based on the classification scheme explained in Section 2. More in detail, Section 5 describes all papers proposing specific IoT application scenarios where LiFi is implemented. Papers analyzed in Section 6 describe different aspects of LiFi integration with other technologies. A physical layer analysis of LiFi for IoT is carried out in the papers analyzed in Section 7. Energy efficiency is the theme tackled in Section 8. Section 9 groups all the papers discussing communication schemes that include LiFi. Contributions proposing positioning algorithms for LiFi are discussed in Section 10.

Furthermore, in Section 11 the most relevant open issues are addressed, also providing some indications of the possible future research directions that the research work can follow in this field. Finally, Section 12 concludes this work.

2. The rationale of the proposed papers taxonomy on LiFi for IoT

The goal of this section is to explain the rationale at the basis of the classification of the papers that apply the multi-faceted aspects of the LiFi technology applied to the IoT environment. The classification approach proposed in this survey follows the taxonomy illustrated in Fig. 1. Specifically, papers are classified and discussed according to the following topics:

- LiFi in IoT applications. This section comprises all the works dealing with the implementation of the LiFi technology in a specific IoT application scenarios. There are some papers that provide only an overview of LiFi in different applications scenarios more suitable for the IoT market [26–28]. Other papers instead discuss a single IoT class of applications where LiFi can be fruitfully employed, like the e-health [29], telemedicine [30], mining environments [31], public transportation systems [32], smart factories [25], appliance automation [33], smart home [6], indoor positioning applications [34, 35], audio applications [36] and power grids [37].
- Integration of heterogeneous communication technologies. This section groups together all the papers that propose hybrid networks in which LiFi is integrated with other technologies like Wireless Fidelity (WiFi) [27, 38–44] or last-generation cellular networks [25, 45, 46]. Different aspects of the integration are faced, starting from a review of the integration in some application scenarios (often discussing the related advantages, limitations and challenges) [25–27, 45, 46], to the implementation of QoSaware strategies [40, 41, 44], proposals taking into account energy efficiency [38, 44], security aspects [39], or procedures at link layer [43].
- **Physical layer analysis.** Papers grouped in this section discuss several aspects related to the physical level of data transmission for LiFi systems in the IoT. Physical layer analysis is important to manage and improve several aspects of LiFi transmission, i.e., energy consumption [17, 18] and data rate [47] through Multiple-Input Multiple-Output (MIMO) techniques [18, 48] and modulation schemes [15, 26, 27, 42]. Also some security aspects related to LiFi can be analyzed at the physical layer [49].
- Energy efficiency. Papers belonging to this section analyze the energy consumption issue for the LiFi technology applied to IoT. The minimization of the energy consumption is achieved through different optimization strategies, ranging from the network design (i.e., the number of Access Points (APs)) [50], followed by the optimal selection of the LiFi AP that satisfies specific energy constraints [44], to the optimal allocation of the UL and/or DL transmission power [17, 44], the selection of the access scheme that minimizes energy consumption [15, 38], or exploiting MIMO transmission [18].

- Design of communication schemes. This section groups together all the contributions focused on the design and analysis of communication schemes and architectures that can be applied in the IoT context, and that implement the LiFi technology. In general, the architectural schemes discussed in this section describe the structure of the transmitter, the receiver, the components adopted to exchange data, and how they are interconnected. The architectural models related to the communication systems can be explicitly designed for specific application scenarios [15, 33, 37, 51], or to generic environments [34, 38, 52–55].
- **Positioning algorithms.** Papers that analyze this topic show how the LiFi technology can be effectively employed to estimate with high accuracy the position of IoT nodes in indoor environments. This task is accomplished through the proposal and implementation of algorithms that estimate the position of the IoT nodes by means of spatial coordinates [35] or time-based synchronization information [56, 57].

At the end of each section dedicated to a specific topic, a detailed analysis of the main findings is given, together with the related lessons learned.

Several papers analyze more than one topic among the ones described above. Thus, for ease of completeness, these papers have been analyzed in more sections of this survey, describing them from the point of view of the different topics they discuss.

3. Related survey and review articles

The key articles related to the subject of this survey are briefly analyzed in this section, along with their main differences concerning the present contribution.

There are two survey papers dealing with LiFi networks [39, 49]. The contribution [39] surveys the state-of-the-art related to authentication and handover protocols in hybrid LiFi/WiFi networks. Specifically, the main security issues and the related solutions are addressed in this analysis. Also handover protocols for hybrid LiFi/WiFi networks are discussed. Unfortunately, this survey is almost totally devoted to LiFi networks in general, without considering explicitly the IoT environment (it is only briefly mentioned in the Conclusions). The survey paper [49], instead, is mainly focused on the physical layer of VLC transmission. In this context, a security-related analysis is carried out, taking into account several factors like channel models, input distributions, network configurations, precoding/signaling strategies, secrecy capacity, and information rates. A discussion of possible research directions related to security for VLC is also carried out. The main drawback of this contribution is that the LiFi technology is only superficially covered as a particularization of VLC and only in a specific scenario of hybrid LiFi/WiFi networks. Furthermore, the IoT environment is not analyzed in detail. Rather, it is only mentioned in relation to the (high) number of connected devices.

Some contributions discuss several aspects of LiFi-based networks for the IoT [1, 6, 26, 27, 30, 45, 49, 58–60]. In all these



Figure 1: Paper taxonomy for LiFi in IoT environments

works, the review of the state-of-the-art literature on this theme is necessarily not thorough, being tailored to justify concepts and propose ideas, describe the implementation of ad-hoc systems, or clarify characteristics peculiar to LiFi. So, differently from the present contribution, they cannot be considered survey papers since the review work is partial and considers only some aspects of the integration of LiFi into IoT.

Papers [26, 27, 45] provide an overview of the LiFi technology, analyzing some technical aspects related to the physical layer (modulation techniques, channel capacity, adopted spectrum, etc.) and proposing application scenarios where LiFi can be employed. A discussion of the integration with other technologies like WiFi [26, 27, 61, 62] or Fifth Generation (5G) [45] is also carried out. In all these contributions, being LiFi the central topic of both these works IoT is only cited and its description neglected. Another intriguing paper is [1], that highlights the main differences between VLC and LiFi. According to this, the IoT environment is neglected, and the review work is focused on the research topics relevant to LiFi, with reference to modulation techniques, physical components, multiplexed access in the channel, network models, interferences and models of integrated LiFi/WiFi networks.

Papers [6, 58] are more related to the IoT world. They dis-

cuss the most relevant challenges and opportunities deriving from the integration of LiFi in IoT. In this context, the contribution [6] performs an exhaustive analysis of advantages and limitations of LiFi in IoT systems. A list of applications that support LiFi in IoT scenarios is also presented. Finally, an application example on smart home automation is designed, that integrates IoT and wireless communication. This paper exhaustively discusses the advantages and disadvantages of such integration, but it mainly presents the idea (as position paper), without analyzing the state-of-the-art literature in support of it. The same topic is tackled in [58], where some solutions are discussed on the adoption of LiFi in IoT scenarios. The goal of this contribution is to evaluate the integration of LiFi and IoT into real-world applications and scenarios. Different aspects are discussed, ranging from use cases to algorithms at different layers of the protocol stack, mobility, standardization activities, and prototypes. The solutions discussed in this paper are developed in the framework of an innovation action project on LiFi, whose focus is on challenges and solutions that allow concretizing the research work on this topic into real-world applications. Like the work [6], this paper does not survey all the state-of-the-art literature on LiFi for IoT.

The approach followed in [30, 59, 60] is to analyze some technological aspects of LiFi systems conceived for IoT applications. In [59], the main features needed to adapt LiFi systems to IoT applications are discussed, with a specific reference to the physical layer. In this layer, several features are analyzed, ranging from connectivity requirements (at different layers of the stack) to mobility support, implementation of integrated wireless/wired networks, and the management of interference. A broadband communication approach at the physical layer is also proposed, that integrates the LiFi technology with Plastic Optical Fiber (POF) links. The study is carried out in the framework of an EU H2020 project on the LiFi development for IoT.

The paper [60] analyzes the scientific contributions related to OWC technologies, including LiFi, devoted to IoT solutions. The review of the literature is more focused on OWC solutions than on LiFi, which is analyzed as part of the OWC environment. The review of the contributions is tailored specifically for 5G and IoT scenarios.

The most significant open issues for LiFi systems as related to the 5G technology are also reviewed in [30]. The review work conducted in this paper touches different aspects ranging from link design to system requirements, challenges and techniques to mitigate the impairments. A prototype of a LiFi system is identified, with a detailed analysis of the structure of the transmitter and receiver. An implementation example based on telemedicine is also considered.

Papers [34, 63] provide an overview of some concepts of LiFi networks. Also in this case, the literature review is aimed at clarifying some conceptual aspects of LiFi technology, without conducting a wide-ranging and detailed description of its integration in the IoT paradigm, which instead is the goal of the present contribution.

The work [34] proposes a system in which LiFi serves IoT nodes. Indoor application scenarios are considered in this work, with a description of the LiFi Access Point (AP) and IoT devices. The proposed system is designed explicitly for LiFi-for-IoT scenarios, describing all the system parts and system architecture and performing feasibility analysis, with particular reference to positioning and power delivery aspects. Specifically, it suggests viable solutions for IoT nodes that only need to deliver data intermittently, such as On-Off Keying (OOK), Pulse Position Modulation (PPM), spatial PPM, and random number modulation by exploiting off-the-shelf LEDs as an enabler for the IoT environment Finally, the main research directions of LiFi for IoT are discussed.

A more general review of LiFi networks is carried out in [63], without any discussion of the integration with the IoT environment, which is relegated mainly to the application area. In this work, the main concepts of LiFi technology are discussed. They encompass LiFi networks, state of the art in standardization activities, and applications. The main features of terahertz communications are also presented for the integration of LiFi into future 6G scenarios.

4. Contribution of this survey and main differences with other surveys and review papers

To the best of the authors' knowledge, the present contribution is the only one providing a wide-ranging and integrated discussion of all the relevant aspects concerning the integration of LiFi in the IoT environment. From the analysis of the most recent literature on this theme, the following topics have been analyzed in this work:

- Coexistence of LiFi and IoT in a specific application scenarios (e.g. e-health, hazardous environments, public transportation systems, home automation, etc.);
- Integration of LiFi and IoT with other communication technologies (like WiFi and optical fibers).
- Physical layer analysis, which extends the survey work made in [49] (that nonetheless mainly discusses security issues). The analysis at this level concerns mostly modulation and coding schema.
- Energy-efficient strategies, aiming to optimize the energy consumption of LiFi systems that can operate in the IoT context. In some contributions, energy efficiency is considered as a QoS requirement.
- Communication schemes, discussing in detail the system components and their interconnections for data transmission/reception in LiFi-based systems. In some works, this topic is related to the development of specific application scenarios.
- Positioning strategies based on LiFi, aiming to detect the position of IoT devices as precisely as possible, and in real-time.

Starting from this schematization followed in this contribution, and comparing the topics listed above with the survey and review papers described in Section 3, the following important differences arise:

- The two survey papers [39, 49] are the only ones that carry out a thorough review of the literature on LiFi technology. Nevertheless, this analysis is not explicitly referred to its integration with the IoT environment. Furthermore, the survey works are mainly dedicated to LiFi security aspects [39, 49] and protocols [39], and on the integration of LiFi with WiFi, neglecting several other relevant topics (i.e.: application scenarios, integration among different technologies, energy-efficient strategies, etc., as listed above).
- The contributions [1, 6, 26, 27, 30, 45, 58–60] cannot be properly considered as survey works, as specified above, since their review work is tailored to the investigation of specific aspects, or to better support the analysis and implementation of particularized schema, thus considers only some aspects of the integration of LiFi into IoT.
- Papers [34, 63] are even more general in their literature review, which is performed to overview some basic concepts of the LiFi technology and not always referred to its implementation in the IoT environment. Also in this case, the analysis of the state-of-the-art literature does not take into account, point by point, all the most important aspects of the application of LiFi to the IoT.

For ease of completeness, Table 1 shows the main differences between this survey work and the other survey and review papers on the adoption of LiFi for the IoT found in literature, highlighting the missing topics of the latter that are covered in this contribution.

5. LiFi in IoT applications

This section describes all the papers that focus on specific IoT applications where the LiFi technology can be effectively adopted. According to the state-of-the-art literature, the IoT applications scenarios chosen for the implementation of LiFi are mainly indoor (even if there are some exceptions to this, as shown in [26, 28, 32, 35]) and range from e-health [29, 30] to harsh environments [31], smart factories [25], home and office [6, 28], industry [28, 37], navigation systems [35] and transmission of audio/voice data [35, 36]. Many of these scenarios are briefly cited in the review paper [26, 27], which also mentions other interesting scenarios like underwater, aircraft, defense, disaster management and risky environments.

Several application scenarios are grouped, listed and briefly discussed in the review papers [26, 27]. The goal here is to better highlight the potential of the LiFi technology in practical application examples, in a more general framework of a technological overview of LiFi and the description of its potential in several scenarios of interest. Many LiFi applications are also suitable for IoT networks, even if this technology is not explicitly described in these works.

The contribution [28] focuses on the integration of LiFi use cases in the IoT environment. Starting from the assumption



Figure 2: Figure on LiFi in indoor and outdoor environments.

that LiFi solutions are usually customized for their own ecosystem, the goal of this study is to propose and discuss a LiFi system concept and how it can satisfy the requirements imposed by IoT applications. The main LiFi use cases, with related requirements for IoT applications, are described pointing out the flexibility in the adoption and versatility of hardware and software components.

The integration of LiFi in IoT for the e-health scenario is analyzed in [29]. The integration between these two technologies is exploited for the enhancement of a health monitoring system, where a doctor can quickly update the health conditions of his/her patients in the cloud. The cloud is the IoT system while the LiFi network guarantees very fast and noninterfering connectivity. the doctor can thus analyze the patient's data and provide real-time feedback to the assisting person. The keyelements of the health monitoring system are precisely identified, and their interaction is described, also providing flow diagrams of the system.

LiFi is also used in mining environments, as described in [31]. This work focuses on critical environments where the management of emergency information is very difficult because of the poor coverage of WiFi technology. Specifically, LiFi is used to monitor the critical conditions detected through IoT sensors. A decision-making system is also developed to detect abnormal conditions in the presence of hazardous gas. Performance analysis of the proposed system is evaluated in a real testbed, describing the hardware components adopted.

The LiFi technology is exploited in [32] for local advertising in the public transportation scenario. This paper focuses on the outdoor scenario (as shown in Fig. 2), where LiFi is used to transmit the local advertisement to devices close to LED sources, but the proposed model is claimed to be suitable also for indoor IoT communications, increasing their data reliability and solving bandwidth bottlenecks. Both transmitter and receiver sections are analyzed in detail for LiFi communication, providing a hardware-based implementation model and testing its performance in terms of signal attenuation and bit error rate.

LiFi is employed to support smart factories in combination with the 5G network in [25]. The main requirements for this use case are analyzed, in terms of coverage, throughput, network infrastructure, data reliability and delay. A factory demonstra-

Covered topics	[49]	[39]	[27]	[26]	[45]	[1]	[6]	[58]	[59]	[60]	[30]	[34]	[63]	This work
Application scenarios			1	1			✓	✓			1	✓	1	1
Energy-efficient strategies														1
Integration of different technologies		1	1	1	1	✓							1	1
Physical layer analysis	1		1	1	1				1	1			1	1
Communication schemes						✓			1	1	1	 Image: A start of the start of		1
Positioning evaluation strategies														1

Table 1: Review of other surveys/review papers and comparison with this survey.

tor that considers the factory environmental conditions and the integration between LiFi (in the access network) and 5G (in the core network) components and protocol stacks is then implemented. Its goal is to demonstrate reliable communication between end IoT devices and application servers.

In the work [33], the LiFi technology is adopted to build an appliance automation model in combination with the Message Queue Telemetry Transport (MQTT) protocol, which is an application layer protocol typical of IoT environments. The proposed model is implemented in a hierarchical hybrid star-tree IoT network, with three levels of hierarchy. The leaf nodes lying at the third level of the hierarchy (the most peripheral one) are provided with LiFi technology and transmit data to a centralized server by using MQTT. The amount of data collected by LiFi nodes is analyzed by means of Machine Learning (ML) algorithms, to predict their temporal behaviour.

The main goal of the contributions [6, 30, 34] is to discuss the main challenges, opportunities [6, 30] and possible research directions [34] arising when LiFi is adopted to enhance the connectivity of IoT nodes. Nevertheless, in all these papers application examples are also presented, to provide a more clear discussion of the covered topics. Specifically, a home automation system integrating LiFi and IoT is discussed in [6]. The focus here is on the design and implementation of a home controlling and monitoring system. The main system functions are described, and a block diagram of the system is proposed, where all the components are identified and described also in terms of network connectivity. An android-based app for appliance control is also proposed, for the practical implementation of the home control system. In the review paper [30], telemedicine is adopted as an application example to address the challenges to be faced when LiFi is adopted in IoT networks. To this end, a LiFi architecture is proposed as a possible implementation scenario within hospitals. The application consists of a circuit transmitting information (an audio signal, in the example) via LiFi. All the components of the prototype are chosen and connected, and the signal is transmitted through LEDs with high illumination capabilities. The main challenges observed in the prototype implementation are also discussed. A more general indoor application scenario is proposed in [34], where LiFi serves IoT nodes. The proposed application scenario is suitable to provide some kind of services such as precise positioning, energy harvesting and security [34]. The system is described in detail, together with all its components, also providing a feasibility analysis in terms of power, energy efficiency, response time and data rate. A comparison with the adoption of LiFi with RF transmission is also carried out, highlighting the main

limitations and challenges brought by LiFi. Finally, the main research directions to enable the LiFi/IoT integration are addressed.

A prototype of a LiFi system transmitting an analog audio signal is also discussed in [36] for IoT applications. The audio signal comes from a mobile phone. It is modulated and converted into light emitted by blinking LEDs. A solar cell receives the light signal and forwards it to a speaker. The system is implemented with real HW components, testing its effectiveness and addressing the main advantages in the adoption of LiFi for applications in IoT.

Location-based applications are discussed in [35], that employ LiFi as integrated with WiFi for indoor scenarios where GPS-based systems obtain poor coverage performance. In addition, a voice-based input system is implemented, that generates voice alerts in presence of hurdles for visually impaired people.

Integration of LiFi in power grids, which is a typical IoT application, is the topic of [37]. The paper discusses the socalled Optical Internet-of-things (OIoT), which is the IoT that adopts optical wireless connections. This concept is translated into power grid environments. This paper analyzes the optical theory on the basis of OIoT. It is then exploited to develop an analytical method for the OIoT performance evaluation, and design the block-based diagram of an optical wireless communication system for the power grid. Finally, different application scenarios in the power grid domain are described, where OIoT can be effectively exploited.

5.1. Lessons learned on application scenarios for LiFi in IoT

The papers analyzed in this section testify that the LiFi technology can be effectively adopted in a wide variety of application scenarios peculiar to the IoT world, where each application has its own traffic features [34]. In these contexts, LiFi can be very advantageous for different reasons: high energy efficiency [32], security [6, 30, 32], high precision in localization [34], huge available bandwidth [6, 29], low-cost components [29], robustness towards electromagnetic interference [6, 28, 30, 32] and capability to serve a huge amount of IoT nodes [32, 34]. Nevertheless, the abovementioned studies suggest that LiFi is still not mature to totally replace the WiFi connections currently adopted in IoT, at least in the short term. Instead, it should be integrated and harmonized with existing technologies [30]. In fact, LiFi suffers from some limitations that can be relevant in some IoT environments. Just to cite some of them: the necessity of a Line of Sight (LoS) and a very short distance between two LiFi nodes, a constant light source to access the network, the high difficulty in implementing LiFi connections in outdoor



Figure 3: Example of Integration of heterogeneous communication technologies.

scenarios due to other interfering light sources (like the sun) and varying weather conditions, and the time and expensiveness needed to set-up and install LiFi infrastructures [6]. As a consequence, LiFi will be worth being adopted in all that IoT applications, especially indoor, that need to be optimized for high throughput, high reliability, or low latency, at the same time optimizing costs [28]. From the analysis carried out in these papers, it is also noteworthy that most probably LiFi will be the best candidate technology for the future IoT [28, 34].

6. Integration of heterogeneous communication technologies

This section describes the proposals for the integration of different communication technologies that include the adoption of LiFi in IoT environments, as shown in Fig. 3. In particular, relying solely on traditional networks to accommodate the explosive growth in bandwidth demand from the burgeoning number of IoT devices is no longer sufficient. In fact, LiFi systems have drawbacks due to the blockages of the light path that limit the received signal due to opaque objects and obstacles. Consequently, combining LiFi and RF can be an easier way to fulfill future IoT requirements. The most widely discussed kind of integration is between LiFi and WiFi communication technologies, as testified by several papers found on this topic [26, 27, 38–41, 43, 44].

The contribution [38] proposes a technique for energy harvesting, in a scenario that combines both WiFi and LiFi. The goal of the proposal is to achieve higher data rates by integrating both technologies rather than using them separately. The analysis is carried out by designing a model that accounts for the capability of the energy-harvesting sensor to transfer the harvested energy to the sensor node or storage charge and to manage the output of sensors. The performance comparison, in terms of data rate and Bit Error Ratio (BER), is conducted for LiFi only, WiFi only, and hybrid scenarios considering bidirectional communication.

Papers [26, 27, 39] discuss the integration between LiFi and WiFi. In these papers, an explicit description of the IoT environment is missing. Nevertheless, there are some aspects de-

tailed that are suitable for IoT-related networks (e.g., some application scenarios, some system specifications, etc.). Specifically, the study [39] attempts to highlight potential security issues when employing LiFi solely and during the handover protocols between WiFi and LiFi technologies. It explores a wide range of LiFi and handover LiFi/WiFi protocols by disclosing security flaws that might lead to severe attacks. It describes some methodologies to increase security, like the adoption of beamforming methods that reduce the Signal-to-Noise Ratio (SNR) of unauthorized entities in VLC networks to zero, or channel matrices for precoded signals that are invertible only for authorized users. Channel security during the handover process is improved by avoiding the share of the same passwords.

Papers [26, 27] basically adopt the same approach to provide an overview of the main features, developments, advantages and challenges of LiFi. More in detail, the contribution [26] proposes an analysis of the LiFi technology, its main developments, its advantages and limitations. The combination of LiFi and WiFi is discussed, with a focus on the advantages of the resulting hybrid network, especially in terms of security, data rate, coverage and precise positioning in indoor environments. Different possibilities of combination are also presented. Another contribution that provides an overview of the implementation of LiFi over WiFi is found in [27]. Starting from the description of the evolution of the VLC technology into LiFi, a technical comparison between LiFi and WiFi is performed. Some specific aspects are mentioned, i.e., IEEE standardization, the implemented topologies of LiFi and WiFi networks, the operation bands, the coverage and the data rate. Also in this paper, these topics are summarized without going into deep detail.

Also the works [40, 41, 44] deal with hybrid LiFi/WiFi IoT networks. In this scenario, the contributions [40, 41] define the optimization problem of the selection of the most suitable AP based on client-specific QoS constraints in the indoor LiFi/WiFi environment. According to the QoS provisioning mechanism, the IoT nodes can select a LiFi or a WiFi AP, given specific QoS constraints. More specifically, the work in [40] presents an innovative access device selection strategy that attempts to optimize the channel quality (i.e., throughput, energy consumption, and delay) while allowing each IoT node to define its subset of QoS requirements. The goal is the reduction of the average delay experienced by each IoT node while satisfying all the QoS constraints. This methodology is implemented in a Mininet-based indoor hybrid LiFi/WiFi network.

The study carried out in [40] has then been extended in [41] by presenting an enhanced algorithm of access device selection that is solved with a game-theory approach. This paper analyses in-depth the LiFi channel while assessing its data rate capacity by varying the angle of inclination of the APs with regard to IoT nodes. It reduces the average delay and increases the average throughput of the network while meeting the QoS requirements of the IoT nodes. The performance of the proposed algorithm is evaluated by emulating a real-world scenario.

In the paper [44], the integration between LiFi and WiFi translates into modeling differently the LiFi and WiFi channels and integrating these two models for the formulation of the optimization algorithm that allows to select the most suitable AP

that maximizes energy efficiency, thus respecting specific QoS guarantees. The proposed methodology is also compared with WiFi-only and LiFi-only scenarios.

The LiFi/WiFi combination is analyzed in [43] at the single interface level. This work considers a virtual interface, composed of the aggregation of two physical interfaces, one based on LiFi and the second on WiFi, obtained by means of a bonding driver at the data link layer. Some performance results are presented, focused on the evaluation of the connectivity downtime occurring during the switching from one physical interface to the other, due to different reasons (signal loss, interface failure, load balancing, etc.)

Other works present an overview of an hybrid LiFi/WiFi network able to overcome the shortcomings of both standalone technologies by combining the fast data transfer capabilities of LiFi with the wide coverage of WiFi, without focusing on the IoT domain. Specifically, [61] presents an overview of LiFi technology, practical concepts for implementing LiFibased indoor networks, and criteria for measuring network performance. It also discusses how to design and set up a LiFibased indoor network as standalone LiFi or hybrid LiFi/WiFi network. Furthermore, [62] presents network architectures, cell deployments, multiple access and modulation schemes, illumination requirements, and backhaul. Moreover, key performance metrics are then reviewed to demonstrate the superiority of hybrid LiFi/WiFi network against standalone networks. Then, the challenges are elaborated on key research topics including user behavior modeling, interference management, handover, and load balancing. It also examines the potential benefits of LiFi/WiFi network for application services, such as indoor positioning and physical layer security.

Another topic on this theme debated in the modern literature consist of the integration of LiFi with 3GPP-based last-generation cellular networks like 5G and beyond, which are suitable for IoT support. This aspect is emphasized in [25, 45, 46].

In this context, in [25] LiFi is integrated in the 5G network. The focus of this work is to exploit the LiFi technology in the access network while using the 5G technology in the core network. Since, differently from 5G, LiFi is a non-3GPP technology, the effort of this study is devoted to the integration of the LiFi and 5G components and the related protocol stacks. A demonstrator is also developed, that shows the reliability of the communication between IoT devices and application servers.

The integration between LiFi and 5G is discussed in [45]. Even though this paper does not propose any novel methodology, it carries out a comparison between the main LiFi performance parameters with the correspondent 5G ones, to evaluate the possibility of implementation of LiFi-based cellular networks. Accordingly, the main aspects of LiFi technology are discussed exhaustively, also briefly mentioning the integration of LiFi with WiFi.

More tailored to the integration of LiFi into future 5G and beyond network is the work [46], which recommends the further integration of 6G in a LiFi/WiFi network to provide high coverage also on subways, aircraft, and trains. Specifically, 6G would ensure connectivity to the public network, LiFi would achieve a reasonable indoor data rate, and WiFi would compensate for LiFi shortcomings (e.g., penetrating walls).

6.1. Lessons learned on the integration of different communication technologies

The analysis of the literature on this topic suggests that the integration process of the LiFi technology with other existing and consolidated technologies like WiFi or 5G is a central topic in the framework of the LiFi technology. In fact, when a new promising technology (like LiFi is) arises, it needs a more or less long period of integration and coexistence with the other existing ones, that have already been implemented in existing network infrastructures. So, it is not surprising that there is a consistent amount of research work that tackles the integration between LiFi and WiFi, which is widely adopted in IoT scenarios. Also the integration with 5G and beyond systems is a promising line of research, since the last-generation cellular networks are standardized also taking into account the IoT environment. As a consequence of this, a relevant effort should be spent on the interoperability among the different standards, developed for architectures, devices, and protocols that characterize WiFi, 5G (and beyond) and LiFi systems.

In the context of this integration, efforts have been devoted to designing communication schemes satisfying specific QoS guarantees. The fulfillment of this goal for LiFi-based networks helps improve the LiFi network performance in terms of different metrics like energy consumption, data rate, and network throughput. On the contrary, this is not a trivial task. Indeed, strategies in this direction always translate into finding the solution to a constrained optimization problem, which is usually a complex and computationally intensive task, not suitable for IoT devices that usually are resource-constrained. The task becomes even more complex when the QoS guarantees are heterogeneous, depending on the features of the IoT nodes.

7. Physical layer analysis

In this section, the strategies related to the physical layer of LiFi transmission in IoT are discussed in detail.

Physical layer aspects can be found in the framework of energy efficiency strategies, as testified by [17, 18]. More specifically, in [17] the enhancement in energy efficiency performance in a LiFi environment for IoT is studied by modeling a Non-Orthogonal Multiple Access (NOMA) channel. This channel is modeled with an optimal power allocation method to achieve greater performance than a standard Orthogonal Multiple Access (OMA) channel while still meeting all of the QoS requirements of a IoT node. Also in paper [18], the energy efficiency of a VLC communication is boosted by leveraging an adaptive MIMO strategy. This paper investigates the effectiveness of various MIMO approaches by identifying the most energy-efficient method while fulfilling an acceptable error rate.

Related to MIMO techniques is also the work [48]. It proposes a MIMO technique for a OOK modulation suited for the VLC. Even though the strategy is presented for only one transmitter and one receiver (being not suitable for IoT scenarios

with high numbers of nodes), the proposed methodology can improve the LiFi communication if compared to conventional schemes, in terms of data rate and communication range.

Many papers discuss modulation schemes for LiFi [15, 16, 26, 27, 42]. In [16], the main modulation schemes are grouped into two different categories, such as single carrier modulated transmission and multiple carriers modulated transmission. The former includes both Pulse-Based Modulation (PBM) (i.e., OOK, Pulse Amplitude Modulation (PAM), Pulse Width Modulation (PWM), PPM [64]) and Continuous Wave Modulation (CWM). In detail, a PBM imposes that the signal is composed of a periodic pulse (AC current) that is added to the illumination (DC current) by modulating the amplitude of the light. On the contrary, the CWM modulates the light intensity with a sinusoidal carrier signal. Herein, the information can be coded either by amplitude (ASK), phase (PSK) or frequency (FSK) shift keying. Finally, if the signal is modulated both in amplitude and phase is called Quadrature Amplitude Modulation (QAM) [9]. The latter uses many carriers to convey information. Specifically, the main method is the Orthogonal Frequency-Division Multiplexing (OFDM) which maps bits with a given modulation scheme (e.g., QAM) by exploiting orthogonal subcarriers. Then, the output is obtained using the Fourier transform [65]. Nevertheless, OFDM signals should be shaped for Li-Fi systems to allow for LED source intensity modulation. Hence, a positive limitation is essential for an optical modulation system to cut off noise and ensure the accuracy of the modulated signal. Indeed, to counteract the bipolar nature of the OFDM signal, the DC-biased optical OFDM (DCO-OFDM) is commonly employed. Alternatively, the asymmetrically clipped optical OFDM (ACO-OFDM) only transfers data on the odd subcarriers. The resulting bipolar signal at the output of the IFFT is clipped at zero to give a non-negative signal. However, it sacrifices some spectral performance to ensure that no data is lost. Finally, even though the DCO-OFDM attain higher spectral efficiency and data rates [61, 62], ACO-OFDM is employed in more energy-efficient applications [62, 66, 67].

Modulation schemes for LiFi are reviewed also in papers [27, 42]. Both these contributions describe the modulation techniques adopted for LiFi, subdividing them into two main groups: modulation schemes in common with radio wave communications, further classified into single carrier and multicarrier modulations, and modulations techniques that are exclusively adopted in LiFi transmission since they exploit features peculiar of light signals. In [15] the coexistence issues among low-power IoT nodes with low data rates and LiFi nodes that can easily manage high data rates and high reliability is faced at the physical layer. In fact, this work evaluates the suitable modulation techniques such as QAM, PAM, Color-Shift Keying (CSK), Discrete Hartley Transform (DHT), DCO-OFDM, interleaved subcarrier mapping, modified data sequence and multi-access techniques such as Wavelength Division Multiplexing (WDM) and Orthogonal Frequency-Division Multiple Access (OFDMA). In addition, the performance of each technique is evaluated in terms of Peak to Average Power Ratio (PAPR), delay and throughput. Also the contribution [26] evaluates the performance of alternative modulation schemes, including OFDM, Filter Bank Multi-Carrier (FBMC), and Universal Filtered Multi-Carrier (UFMC) in a LiFi environment using MATLAB simulations. More in detail, this work emphasizes the benefits of using UFMC with respect to OFDM in terms of BER, PAPR, spectral density, and spectral efficiency while at the same time guaranteeing a lower implementation complexity.

Different physical layer aspects are tackled in [49]. This contribution highlights the security challenges that might arise while using VLC communication technology by examining several configurations for data transmission such as Single-Input Single-Output (SISO), Multiple-Input Single-Output (MISO), MIMO, and hybrid LiFi/WiFi systems. To cope with this issue, this paper reviews different channel models, user mobility, and Key Performance Indicators (KPIs) capable of monitoring the safety of this technology. Moreover, security approaches at the physical layer are investigated for boosting VLC channel security, also analyzing the impact of factors like input signaling, the number of eavesdroppers, and Channel State Information (CSI) availability at transmitting nodes on the security performance of the system.

Another contribution is found in [47], that deals with the parallel transmission of data in a LiFi network. This work considers an area covered by several LiFi APs simultaneously connected to an IoT node. In this scenario, an optimal resource allocation algorithm is proposed, that aims to maximize the achievable data rate of the LiFi network. Several factors are taken into account in the resulting optimization problem: the channel model, the node mobility, and the light-path blockage modeled by analytical functions.

7.1. Lessons learned on physical layer strategies in LiFi transmission

This section explores papers that discuss the implementation of traditional RF modulation schemes in the optical domain, including the modifications and adaptations needed to make them suitable for use in a LiFi network (such as adding a DC-bias). However, it is not yet proven whether all of these approaches adopted for the RF domain can be effectively implemented in their current form also in LiFi networks, without introducing penalties like a reduction in spectral efficiency and data rate. Furthermore, a recurring theme in these works is the possibility to consider a LiFi/WiFi network, which still allows for the use of traditional RF approaches by trying to fit the well-known results into the optical domain. Unfortunately, none of the current state-of-the-art approaches precisely focuses on the IoT environment whenever addressing these considerations.

Papers dealing with this topic reveal that the almost totality of the physical layer aspects are well-known concepts, related to modulation schema and multi-antenna transmission techniques. The most relevant aspect is the description of features of the physical signal that are peculiar to LiFi transmission since they exploit the characteristics of the light signal. Results testify that some schemes can outperform the modulation schemes adopted in WiFi networks in some network configurations. Another issue is the coexistence at the physical layer between classical WiFi nodes, which typically are low-power and low-rate nodes, and LiFi nodes that can manage very high data rates. This is another aspect, strictly related to the LiFi/WiFi integration discussed in Section 6, that requires further research efforts.

There are also some security aspects discussed at the physical layer. The starting assumption is that the LiFi signal is, by its own nature, open and broadcast, and this surely poses important security issues also for LiFi networks. Security is faced from the point of view of the physical signal, involving aspects like beamforming, MIMO techniques, modulation and precoding schemes, and CSI.

8. Energy efficiency

This section analyzes all the research papers proposing solutions for energy efficiency suitable for LiFi in IoT scenarios [15, 17, 18, 38, 44, 50]. In all the analyzed papers, energy efficiency may be achieved by:

- selecting the optimal number of APs [50];
- electing the optimal LiFi AP [44];
- properly allocating transmission power in UL and/or DL directions [17, 44];
- choosing the most appropriate multiple access scheme [15, 38];
- designing antenna arrays in MIMO transmission techniques [18].

As testified in [15], at the physical layer an energy-efficient transmission in LiFi networks is influenced by the modulation scheme. Several modulation schemes can be adopted in LiFi systems, as highlighted in Section 7. Nevertheless, the choice of the modulation technique in IoT scenarios, where nodes are power- and resource-constrained, must be decided carefully. Indeed, multi-carrier modulation techniques like OFDM and its variants can provide higher spectral efficiency and throughput; nevertheless, they typically suffer from high computational complexity and energy consumption [34, 68, 69]. So, these techniques are not recommended for IoT nodes. From this point of view, single-carrier and pulse-based modulation techniques like OOK, PAM and CSK could be more effective. Such modulation schemes optimize the optical power utilization, also simplifying the design of the transmitter, where energy saving can be obtained by exploiting OOK. More specifically, higher energy efficiency can be reached if the OOK is implemented with switching transmitters where the power consumed is mainly due to the LED energy consumption [21]. In fact, by properly setting the amplitudes of the on and off states in the OOKmodulated signal, the dissipated energy in the optical transmitter can be made very negligible [21].

One of the main strengths of LiFi technology is its ability to provide both connectivity and illumination, indeed, [50] proposes a trade-off between the number of APs and energy consumption, using constraints that impose a threshold regarding the minimum level of illumination of the environment, by maximizing network performance in terms of the achieved data rate per user. More APs installed in the network mean a greater data rate experienced by each user (but the generated inter-AP interference will also increase). Conversely, the fewer the number of APs, the less energy is consumed (but the average level of illumination will also decrease). The decisive choice is made by the network designer, who selects the most essential metric from network performance and energy usage.

The contribution provided in [44] aims at decreasing the energy of IoT nodes in a hybrid LiFi/WiFi network. Specifically, it designs a Mixed-Integer Linear Programming (MILP) problem that finds the optimum AP while satisfying all QoS constraints, the maximum number of simultaneously connected IoT nodes to each AP, and taking into account the upper bound of each IoT node transmission power. More in detail, the goal is to maximize the energy efficiency of every IoT node, by setting the minimum transmission power able to meet the throughput constraints related to energy consumption. Performance results are compared with WiFi-only or LiFi-only network scenarios.

In the context of multiple access schemes, a promising technique is given by NOMA. Differently from orthogonal multiple access schemes (like OFDM previously discussed) where users are allocated into orthogonal time and frequency resources, in NOMA all the users share simultaneously the whole time and frequency resources, since they are multiplexed in the power domain and demultiplexed by means of Successive Interference Cancellation (SIC) techniques [17, 70]. Some works in literature have proven the effectiveness of this scheme in multi-user IoT scenarios for 5G systems [70-72]. The same contribution [70] also testifies the suitability of NOMA schemes also for VLC networks. It is noteworthy that the adoption of NOMA schemes overcomes the inherent limitations of techniques like the abovementioned OOK. Indeed, even if OOK is one of the simplest modulation techniques adopted in optical communications, it is not always suitable for high-rate transmission due to its low spectral efficiency. Conversely, also NOMA schemes can be made energy-efficient in LiFi systems by designing adhoc power allocation schemes that optimize the power allocated by users, at the same time improving the error performance of the system [70].

Given the considerations reported above, another lowcomplexity power adaptation strategy is proposed in [17] to improve the energy efficiency in a LiFi transmission for IoT environments. The proposal consists of a NOMA scheme that optimally adapts transmission power to maximize the energy efficiency in both DL and UL channels, achieving QoS guarantees for bidirectional communication. To this end, it proposes a closed-form solution with a low-complexity power allocation scheme suitable for LiFi bidirectional channel (i.e., visible light for the DL, infrared for the UL). The network energy efficiency is improved by reducing the likelihood of user outage while meeting all the QoS guarantees.

The lack of transmission methods capable of spatially multiplexing numerous devices without requiring a recurrent fee payment (e.g., NB-IoT) is critical in IoT scenarios. The contribution [15] addresses this issue by providing an innovative energy-efficient communication strategy based on optical OFDM in LiFi environment, that achieves interference-free communication. In the proposed system, LiFi users coexist with IoT devices under a common LiFi AP. This scheme is useful for taking the most appropriate decision on the combination of multiple access and modulation techniques in both the DL and UL channels, to save energy. Delay and throughput performance are also evaluated.

The work [38] provides a model based on the integration of an energy-harvesting wireless sensor network with hybrid LiFi/WiFi communication techniques. This study presents a model that allows for multiple channel accesses while reducing interference. It accounts for the capability of the energy harvesting sensor to transfer the harvested energy to the sensor node or storage charge and to manage the output of sensors. In this context, the single-carrier modulation schemes (e.g., PPM, OOK, PAM) are analyzed for LiFi, also addressing dimming performance for the OOK for visible light systems [38] or in hybrid visible light/infrared communication systems [20]. Multicarrier modulation techniques, like OFDM, are also mentioned as more effective for high data-rate transmission, or in presence of signal distortion and frequency selectivity in the optical channel [38]. It is well known in fact that OFDM presents higher spectral efficiency and robustness against dispersion and varying channel conditions [21]. This contribution also analyzes the CSK as a LiFi-specific modulation. Finally, the simulation results show that a bidirectional multiaccess/multiuser hybrid LiFi/WiFi communication scheme can be effectively utilized due to its ability to provide interoperability for multiple devices/users at a high data throughput rate of 25 Mb/s.

As remarked in [73], MIMO can be effectively adopted in IoT environments due to its capability of serving simultaneously a large amount of IoT nodes, at the same time improving energy and spectral efficiency and increasing the overall system capacity. This holds true especially when a large number of antennas is deployed at the base station side [73]. Several other contributions also highlight the effectiveness of the MIMO adoption in different frequency bands, including millimeter wave (mmWave), terahertz (THz) and optical wireless bands, thus guaranteeing reliable and energy efficient communication in several IoT scenarios [74, 75].

Indeed, different MIMO transmission techniques are proposed in [18] for an indoor LiFi system based on VLC and employed in IoT scenarios. In detail, an adaptive method is proposed to enhance the energy efficiency of MIMO for VLC. Specifically, by varying the channel conditions, the desired spectral efficiency, and a target error rate, the MIMO technique is selected from three options: repetition coding, spatial multiplexing, and a modified version of spatial modulation. Finally, MIMO technique with the lowest energy requirement is chosen based on the user's location, thus reducing energy consumption, by also considering the desired spectral efficiency and target BER. This adaptive method can be particularly useful in IoT applications where energy consumption is a critical concern.

Since MIMO is based on spatial diversity for signal transmission, its performance is influenced by the position of the LiFi network elements (light sources and photo-detectors), and by the channel conditions [76]. The analysis carried out in this paper shows that this diversity factor impacts the main parameters



Figure 4: Example of design of communication scheme.

that contribute to energy consumption, i.e., the optical power needed to reach a target BER. So, the optimization of the energy consumption depends on the chosen MIMO technique as related to the specific IoT scenario (more details on this can be found in [18]).

8.1. Lessons learned on strategies for energy efficiency optimization

The works analyzed in this section suggest that the minimization of energy consumption is of paramount importance in LiFi networks serving IoT systems. Indeed, IoT nodes are usually battery-powered and batteries should last for a very long time. This issue is exacerbated for LiFi access networks, thought to achieve very high data rates that require a higher power consumption at the node side to meet this requirement. Efforts in this direction have been made mostly by analyzing data transmission at the physical layer, considering LiFi channel models, energy-efficient modulation schemes, the adoption of MIMO strategies and the optimization of the transmission power for the LiFi nodes.

9. Design of communication schemes

This section identifies and discusses the contributions that analyze communication schemes applicable to IoT environments adopting the LiFi technology, as shown in Fig. 4.

In this framework, a basic approach is proposed in [54]. In this work, the system is made up of a smartphone and a LED that implements a LiFi bidirectional channel. The former is outfitted with a flashlight as a transmitter and a camera that serves as a receiver. The initial stage in acquiring information is a picture captured by the camera, which extracts the brightness component of the image. The LED position is then detected. The signal is decoded after equalization and quantization. Finally, the information can be successfully collected following the error-checking procedure. Furthermore, the LED can act both as a transmitter and a receiver. Specifically, variations in incoming light intensity can be detected by charging a reversed biased LED and measuring its electric tension after a short time interval [54]. Specific efforts have been made in the design of communication schemes for outdoor scenarios, which are critical for LiFi networks [52, 53]. The authors in [52] propose use cases suitable for a disaster scenario by exploiting the VLC over a lighthouse to communicate with fishermen in an emergency situation. The outdoor LiFi communication scheme is implemented following the DMX512-A standard that enables the unidirectional data transmission for VLC. Specifically, the device block diagram includes a power supply, a serial interface linked directly to a laptop and a microcontroller that manages the light-based transmitter or receiver. Another contribution in [53] designs a Software-Defined Networking (SDN)enabled LiFi attocellular network virtualization able to share the infrastructure over multiple Mobile Virtual Network Operators (MVNOs). This is possible thanks to a resource slicing application offered by the northbound of the SDN controller that is also able to handle the on-demand creation of a LiFi DL channel. Under heterogeneous bursty traffic conditions, the proposed approach aims to maximize the data rate and optimize the allocated resources of each MVNO infrastructure, while minimizing the fees incurred by the providers. To this end, a heuristic matching game method is exploited to properly address the resulting optimization strategy.

Differently from the previous works, communication schemes for indoor scenarios are proposed in [15, 33, 37, 51]. The authors of [15] propose a novel architecture for LiFi-based indoor networking for IoT users. In detail, the work designs a LiFi AP with a transmitter that can operate on three distinct wavelengths (i.e. red, blue, and green) to communicate with different receivers at the same time, and a receiver consisting of a photodetector capable of gathering IoT node signals. Furthermore, it proposes equipping IoT nodes with one element capable of sending in the infrared band and the other component capable of receiving by filtering the system's different wavelengths. Once the communication equipment has been described, the block diagram of the various forms of modulation employed in the DL channel is defined, beginning from modulation and moving through the DAC and ADC until the signal demodulation. The work in [33] envisions the deployment of a one-way LiFi hotspot that is engaged for DL communication only. It transmits the data stream to the end user by generating it via LED. First, the data stream is encoded by the transmitter, enabling an asynchronous communication based on MQTT and with frames as small as 11 bits (one start bit, two stop bits, and eight ASCII characters). Finally, the receiver may decode the broadcast signal into alphanumeric characters by analyzing the received frame with the help of a solar panel that detects tiny changes in light. The benefits of adopting LiFi in high interference conditions due to high voltage and strong magnetic fields are shown in [37]. First, this contribution introduces a novel concept called the Internet of Light, in which lights are assigned an IPV6 address and are considered network nodes. In detail, a sensor is inserted inside a High Voltage Direct Current converter to provide constant feedback on its functioning state. Finally, the voltage is detected by the sensor, which sends the signal created by a modulator to the light source, which is received by an optical antenna of the receiver and demodulated before passing to the signal processing phase. According to the

authors of [51], a LiFi network is used to achieve device-todevice (D2D) communication in an ultra-dense Industrial Internet of Things (IIoT) environment, by adopting a scenario drawn from the IEEE 802.11bb standard for VLC communication.

The choice of such a standard is particularly suitable for the analysis of realistic IIoT scenarios, like the one described in detail in [77], especially for what concerns the channel model. Furthermore, as remarked in [78], if compared to the other optical wireless communications standards like IEEE 802.15.7, IEEE 802.15.13, or the ITU-T G.9991, IEEE 802.11bb introduces only slight modifications to the MAC layer of the IEEE 802.11 standard adopted by for WiFi, thus easing the LiFi/WiFi coexistence which is an important aspect in an IoT environment [38, 44, 62].

In this scenario, the network is split into clusters identifiable by a given LiFi AP, and each of these AP covers a variable number of mobile devices that act as relying upon the AP for the gathered data from the IIoT nodes. Specifically, the IIoT nodes use the measured received optical intensity to choose the optimal mobile device suitable for the D2D. Finally, the network performance is quantified by evaluating different mobile device speeds, transmitter heights from the ground, and room sizes.

Unlike the previous contributions, papers [34, 38] focus on energy harvesting strategies. The work in [34] introduces IoT nodes that are self-sustaining in terms of energy and conducts a feasibility analysis using the existing technology. It specifically provides three alternative setups that employ an AP outfitted with a LED for transmission, a photodetector, and a CMOS camera for data reception. In the first configuration, an IoT node is equipped with a solar panel that can perform both energy harvesting and data reception in DL, while for transmission a liquid crystal shutter is employed, which can perform both backscattering and UL communication in VLC. The second configuration employs a solar panel that is also responsible for waking up the IoT node to allow communication in the UL channel exploiting the infrared technology. The third configuration presents a hybrid node that uses both LiFi and RF, with the DL channel acting as a control channel to efficiently employ the RF, interrupting the sleep state. This study also illustrates three potential autonomous setups from the standpoint of energy, which is a crucial requirement for the IoT.

Another contribution analyzes a typical indoor IoT environment for smart housing and smart industries using a hybrid LiFi/WiFi network [38]. The proposed system includes an EHsensor node that can manage energy harvesting and communicates utilizing the frame structure defined in the 802.15.7 standard. When smart devices want to connect with EH-sensor, they transmit data to the frame creator block. Then, the output is sent to the encoder block. Once encoded, the data frame is sent from the modulation block to the LED driver, which produces blue light to interact with the EH-sensor. Indeed, it has an optical color filter to filter the blue-colored LED light and receive the data frame with the help of the photodetector. Furthermore, it keeps the received data in the buffer after demodulating and decoding the data frames. When EH-sensors wish to connect with smart devices opposite approach is taken utilizing a green LED



Figure 5: Example of a security system.

light to avoid interference.

Finally, the contribution [55] proposes a solution for data storage in a secure LiFi environment for the IoT. Specifically, it describes a LiFi-IoT environment in which the sensors are coupled to a microcontroller capable of driving an array of LiFi transmitters, as shown in Fig. 5. Consequently, another array of photodetectors serves as an AP through the proposed system, which also includes a Advanced Encryption Standard (AES) algorithm adopting 128-bit symmetric keys and a Role-Based Access Control (RBAC) authentication and authorization mechanism. As a result, a method to securely store the sensed data is implemented.

9.1. Lessons learned on the design of communication schemes

The communication schemes proposed in the analyzed works often investigate in depth the components of the communication systems and relate them to specific use case scenarios. An important aspect covered in these contributions is the proposal of communication schemes for outdoor scenarios, that are more critical for LiFi-based communication. In this context, the optimal selection of available resources to maximize the network data rate seems to be an interesting topic, bringing encouraging performance results. Unfortunately, finding the solution to this kind of optimization problem is time- and resource-consuming, making it difficult to meet the real-time requirements that are instead important in some IoT scenarios. Also, some standardization aspects and the coexistence among different modulation schemes at the physical layer are discussed, thus making the proposed schemes suitable for implementation into real testbeds.

It can be noted anyway that the proposed schemes are quite simple, in most cases being made of a single transmitter and a single receiver. More complex, multi-node scenarios, that are much closer to the IoT reality, are almost totally neglected or are evaluated theoretically, or their performance analysis is carried out through simulations. Also, a compliance analysis of the different standards adopted, wherever implemented simultaneously in the proposal, would be beneficial for practical implementations and performance evaluation in real-world scenarios, especially when LiFi is integrated with WiFi technology. Another important lesson learned is that almost all the proposed communication schemes are described at the physical layer. To strengthen the analysis and provide a more exhaustive description of the schemes, an analysis of the schemes at higher layers (e.g., network layer, application layer, etc.) with the related standardized protocols adopted, would be beneficial.

Some security aspects are also tackled in the context of communication schemes. Nevertheless, security protocols for LiFi is still a work in progress that requires more investigation. Actually, the solutions proposed focus mainly on cryptographic mechanisms. Nevertheless, they must be chosen accurately, to meet the resource-constrained nature of IoT nodes, at the same time being fully compliant with the ongoing standardization efforts in the LiFi environment.

10. Positioning algorithms

This section describes the positioning algorithms proposed in LiFi networks for IoT applications. The proposed algorithms allow the localization of IoT nodes, also in hybrid indoor and outdoor environments, as shown in Fig. 6.

The work [35] provides an integration of the outdoor navigation system (e.g., Global Positioning System) with an indoor navigation system (e.g., LiFi technology). It designs a technique for indoor localization, which is primarily used to store a series of points that represent the latitude and longitude while it proposes a triangulation method for outdoor locating systems.

According to [56, 57], the implementation of LiFi positioning algorithms is a promising solution for indoor applications [56] and industrial environments [57]. In both these contributions, an indoor standard-compliant positioning system based on the so-called time-of-flight approach is implemented. It makes use of the frame structure of LiFi transmission, as stated in the International Telecommunication Union (ITU) Telecommunications Sector G.9991 recommendation [79]. Based on the information contained in the LiFi frame, two kinds of time information are derived: coarse time information (obtained by using the frame synchronization preamble) and fine time information (derived from the channel estimation preamble), to estimate the relative distance between each LiFi AP and the IoT node with high precision.

Real-world tests have been used to validate the proposed system.

10.1. Lessons learned on positioning algorithms and localization techniques

Positioning algorithms can be useful to estimate the position of IoT nodes with high accuracy, thus being of great importance in some industrial environments or indoor scenarios where the position of an IoT node must be derived precisely. The main drawback of such techniques resides in the complexity of the LiFi network, which must be composed of a high number of APs to combine the information on the node position coming from the different APs, and the short coverage range of the LiFi signal that is negatively influenced by attenuations and obstacles. Another issue is that the positioning algorithms proposed in the state-of-the-art literature could be energy-consuming, especially in scenarios with high mobility of IoT nodes, since



Figure 6: Example of indoor positioning approach.

their position needs to be continuously updated as they move in the reference area. So, their implementation should be carefully evaluated in such scenarios, when energy saving becomes a relevant issue.

11. Challenges and future research directions

This section aims at addressing the main challenges of LiFi adoption in IoT scenarios as derived from the analysis of the current literature, to find the related missing gaps. Based on this, possible research directions can be drawn to suggest what kind of themes deserve to be further investigated by future research work.

According to the analysis of the current literature, LiFi technology can fulfill the majority of the requirements of future systems put forward in the IoT field, including 5G, 6G, and others. Sharing LiFi and IoT technologies can be a highly effective solution to many of the challenges that take place in WiFi access networks, particularly in terms of data rates, energy consumption, and scalability. This validates the potential of LiFi in a wide variety of IoT scenarios. However, it should be noted that most of the LiFi-based technology is still under investigation. Therefore, in the authors' opinion, it is necessary to pay attention to the obligation to examine crucial theoretical issues.

The first issue is that building LiFi networks to serve IoT systems requires the use of a small cell architecture, which leads to a dense deployment of LEDs. As highlighted in the analyzed papers, this raises the problem of managing inter-cell optical interference, in the framework of light transmission which is different from RF signal transmission. Additional research is needed in this direction. In this case, it is desirable to conduct full-scale experiments using models of real network sections. This aspect is another interesting research direction that can be further explored to provide innovative and effective theoretical models for the LiFi part of the network in IoT scenarios. Another problem that needs further research efforts is the management of the handover of IoT nodes in indoor scenarios with multiple APs and mobile devices. This aspect is important, especially in dense scenarios with real-time data exchange, to minimize service interruptions and maximize system efficiency.

The second issue is related to the thorough evaluation of the throughput performance of IoT systems when using strategies

of interference management like MIMO. Some works coping with this issue propose the adoption of MIMO techniques to increase the system throughput and energy efficiency. Nevertheless, the effectiveness of such adoption is considered as for RF systems. Also in this case, a thorough evaluation of the real performance of MIMO together with its effectiveness in terms of throughput and quality of service would be beneficial in future research works, especially in real testbeds.

Another interesting research field is related to the challenging issue of the proposal and adoption of analytical models for LiFi explicitly tailored for the IoT domain. In fact, many papers in the scientific literature focus on the analytical models for the LiFi network ([11, 80–85]). Nevertheless, all the models proposed in the abovementioned papers refer only to LiFi systems, totally neglecting their integration with the IoT environment. Furthermore, there is still a lack of a reliable channel model for these systems [12]. For instance, unrealistic assumptions are made about the receiver's placement, such as presuming that the LiFi receiver is vertically upward or randomly located within the communication range of the transmitter [12].

Consequently, it is challenging to adapt these models to an environment full of constraints such as the IoT is. Indeed, there is a need to comprehend, design and validate the models produced for LiFi networks in the IoT context. At the same time, mathematical expressions and formulas taken from monographs are used, which do not fully take into account the parameters of modern transmitters and receivers of light signals in the IoT domain. Thus, it would be interesting to create a layout of the minimum configuration of the LiFi system, take measurements and validate the model as accurately as possible by comparing the measurement results with the outputs of the analytical model. This aspect has currently not been properly addressed by the current literature and requires a relevant effort, to properly align analytical results with experimental ones in integrated LiFi/IoT scenarios, and iteratively refine the proposed models accordingly, to solve LiFi network design problems in the IoT domain.

As a result of the conducted survey work, it is evident that much effort in the field of LiFi technology has been devoted to the analysis of the physical layer of light signal propagation between the transmitter (LED) and the photodetector. There are no exhaustive studies of LiFi systems that would analyze the processes of information from the source to the recipient at all levels: physical, channel, network, and above, even by adopting cross-layer approaches. The key challenge is that each level has its own methods and procedures of data flow management, and there are opportunities to improve the efficiency of the system by analyzing the way(s) data is packetized and packets are managed at the different layers of the stack. In this context, also security assumes relevancy. More specifically, the management of authentication and authorization procedures is actually a challenge and needs further research to find the best trade-off among conflicting requirements, such as the implementation of lightweight security procedures to save the computational resources of IoT nodes, the layer(s) of the stack at which these procedures should e implemented, and the maintenance of the security guarantees when handover among different APs occurs. From this point of view, centralized or distributed security procedures could be implemented; nevertheless, they should take into account the specific IoT application scenario and the system architecture. In any case, by using recent achievements in solving the problems of information transfer at each level, it is feasible to identify new potential directions for the development of even more efficient LiFi systems.

12. Conclusions

This work aims at providing a thorough overview of the research activities on the LiFi technology applied to the IoT scenario. To this end, the available scientific literature has been investigated in detail and classified, to identify the covered topics and the current research trends. After a detailed description of the novel aspects of this work and the main differences from the other surveys and review papers, the surveyed papers have been classified based on the different topics, that range from the adoption of LiFi in IoT applications to integrated communication technologies, the solution to optimization problems, QoS provisioning, energy efficiency, physical layer analysis, positioning strategies, security and proposals of communication schemes. Finally, considerations on the main challenges and possible future research directions have been discussed, to provide some precise guidelines that can drive future efforts in this research field.

From the analysis of the surveyed literature, some important conclusions can be drawn. First, there are some topics that attract much interest, like the adoption of LiFi technology in different application scenarios and the integration or coexistence of LiFi with other technologies like WiFi or cellular networks. This is testified by the number of works on these topics. Second, the QoS provisioning is mostly related to the solution to optimization problems, that sometimes is hard to find due to the complexity of the problem itself and not always suitable for IoT devices, which usually have reduced computational resources. Third, the energy saving issue is crucial in IoT environments, where nodes are battery-powered. Nevertheless, for the sake of feasibility in IoT environments, the proposed strategies must keep a low complexity and a high transmission efficiency. Fourth, it is very important to focus the research efforts on the coexistence of LiFi with other communication technologies (WiFi or cellular networks), preventing communication systems and network architectures to be designed from scratch. The main difficulty related to this topic is the integration and interaction among different protocols, developed by different standardization bodies, and at different layers of the stack. Additional efforts have thus to be spent in this direction.

References

- H. Haas, L. Yin, Y. Wang, C. Chen, What is LiFi?, in: 2015 European Conference on Optical Communication (ECOC), 2016, pp. 1–3. doi: 10.1109/EC0C.2015.7341879.
- [2] H. Haas, L. Yin, Y. Wang, C. Chen, What is LiFi?, Journal of Lightwave Technology 34 (6) (2016) 1533–1544. doi:10.1109/JLT.2015. 2510021.

- [3] M. S. Islim, R. X. Ferreira, X. He, E. Xie, S. Videv, S. Viola, S. Watson, N. Bamiedakis, R. V. Penty, I. H. White, A. E. Kelly, E. Gu, H. Haas, M. D. Dawson, Towards 10Gb/s orthogonal frequency division multiplexing-based visible light communication using a GaN violet micro-LED, Photon. Res. 5 (2) (2017) A35–A43. doi:10.1364/PRJ. 5.000A35.
- [4] D. Tsonev, S. Videv, H. Haas, Towards a 100 Gb/s visible light wireless access network, Opt. Express 23 (2) (2015) 1627–1637. doi:10.1364/ 0E.23.001627.
- [5] H. Haas, High-speed wireless networking using visible light, Spie Newsroom 1 (1) (2013) 1–3.
- [6] M. Sathiyanarayanan, V. Govindraj, N. Jahagirdar, Challenges and opportunities of integrating Internet of Things (IoT) and Light Fidelity (LiFi), in: 2017 3rd International Conference on Applied and Theoretical Computing and Communication Technology (iCATccT), 2017, pp. 137–142. doi:10.1109/ICATCCT.2017.8389121.
- [7] L. E. M. Matheus, A. B. Vieira, L. F. M. Vieira, M. A. M. Vieira, O. Gnawali, Visible Light Communication: Concepts, Applications and Challenges, IEEE Communications Surveys & Tutorials 21 (4) (2019) 3204–3237. doi:10.1109/COMST.2019.2913348.
- [8] M. Z. Chowdhury, M. T. Hossan, A. Islam, Y. M. Jang, A Comparative Survey of Optical Wireless Technologies: Architectures and Applications, IEEE Access 6 (2018) 9819–9840. doi:10.1109/ACCESS.2018. 2792419.
- [9] M. S. Islim, H. Haas, Modulation techniques for li-fi, ZTE communications 14 (2) (2019) 29–40.
- [10] E. Ramadhani, G. Mahardika, The technology of lifi: A brief introduction, in: IOP conference series: materials science and engineering, Vol. 325, IOP Publishing, 2018, p. 012013.
- [11] M. D. Soltani, M. A. Arfaoui, I. Tavakkolnia, A. Ghrayeb, M. Safari, C. M. Assi, M. O. Hasna, H. Haas, Bidirectional Optical Spatial Modulation for Mobile Users: Toward a Practical Design for LiFi Systems, IEEE Journal on Selected Areas in Communications 37 (9) (2019) 2069–2086. doi:10.1109/JSAC.2019.2929381.
- [12] M. A. Arfaoui, M. D. Soltani, I. Tavakkolnia, A. Ghrayeb, C. M. Assi, M. Safari, H. Haas, Measurements-Based Channel Models for Indoor LiFi Systems, IEEE Transactions on Wireless Communications 20 (2) (2021) 827–842. doi:10.1109/TWC.2020.3028456.
- [13] D. Ramananda, A. M. Sequeira, S. R. Raikar, C. K. Shanbhag, Design and Implementation of LiFi Communication system, in: IOP Conference Series: Materials Science and Engineering, Vol. 594, IOP Publishing, 2019, p. 012041.
- [14] R. Sharma, D. S. Gurjar, E. Rahman, A. Raghav, P. Shukla, V. Mishra, LiFi Technology: A Breakthrough for Massive Data Rates in Indoor Applications, Springer Nature Singapore, Singapore, 2022, Ch. 11 June 2022, pp. 63–79. doi:10.1007/978-981-19-0770-8_6.
- [15] D. N. Anwar, R. Ahmad, A. Srivastava, Energy-Efficient Coexistence of LiFi Users and Light Enabled IoT Devices, IEEE Transactions on Green Communications and Networking 6 (2) (2022) 930–950. doi:10.1109/ TGCN.2021.3116267.
- [16] A. M. Damerdash, D. Abdelhameed, M. Aly, E. M. Ahmed, M. A. Ahmed, Energy efficiency assessment of power electronic drivers and LED lamps in Li-Fi communication systems, Energy Reports 7 (2021) 7648–7662. doi:https://doi.org/10.1016/j.egyr.2021.10. 112.
- [17] C. Chen, S. Fu, X. Jian, M. Liu, X. Deng, Z. Ding, NOMA for Energy-Efficient LiFi-Enabled Bidirectional IoT Communication, IEEE Transactions on Communications 69 (3) (2021) 1693–1706. doi:10.1109/ TCOMM.2021.3051912.
- [18] I. Tavakkolnia, C. Chen, R. Bian, H. Haas, Energy-Efficient Adaptive MIMO-VLC Technique for Indoor LiFi Applications, in: 2018 25th International Conference on Telecommunications (ICT), 2018, pp. 331–335. doi:10.1109/ICT.2018.8464933.
- [19] Ieee draft standard for local and metropolitan area networks part 15.7: Short-range optical wireless communications, IEEE P802.15.7/D3, August 2018 (2018) 1–412.
- [20] X. You, J. Chen, Y. Zhong, S. Chen, C. Yu, Efficient Dimming Control with Time Domain Hybrid Modulation in Indoor Hybrid Visible Light/Infrared Communication Systems, in: 2019 24th OptoElectronics and Communications Conference (OECC) and 2019 International Conference on Photonics in Switching and Computing (PSC), 2019, pp. 1–3.

doi:10.23919/PS.2019.8817648.

- [21] M. Hinrichs, P. Berenguer, J. Hilt, P. Hellwig, D. Schulz, A. Paraskevopoulos, K. Bober, R. Freund, V. Jungnickel, A Physical Layer for Low Power Optical Wireless Communications, IEEE Transactions on Green Communications and Networking 5 (1) (2021) 4–17. doi:10.1109/TGCN.2020.3038692.
- [22] H. Jmila, G. Blanc, M. R. Shahid, M. Lazrag, A Survey of Smart Home IoT Device Classification Using Machine Learning-Based Network Traffic Analysis, IEEE Access 10 (2022) 97117–97141. doi:10.1109/ ACCESS.2022.3205023.
- [23] K. Lueth, M. Hasan, S. Sinha, S. Annaswamy, P. Wegner, F. Bruegge, M. Kulezak, State of IoT-spring, Tech. rep., Spring, East Lansing, Michigan (May 2022). URL https://iot-analytics.com/

number-connected-iot-devices/

- [24] A. S. Abohamama, A. El-Ghamry, E. Hamouda, Real-Time Task Scheduling Algorithm for IoT-Based Applications in the Cloud–Fog Environment, Journal of Network and Systems Management 30 (4) (2022). doi: 10.1007/s10922-022-09664-6.
- [25] M. Müller, M. Emmelmann, D. Behnke, D. Schulz, K. L. Bober, C. Kottke, V. Jungnickel, T. Metin, LiFi with 5G for the Smart Factory, in: 2022 IEEE Wireless Communications and Networking Conference (WCNC), 2022, pp. 2310–2315. doi:10.1109/WCNC51071.2022.9771969.
- [26] S. Alshami, Performance Analysis of OFDMA, UFMC, and FBMC for Optical Wireless Communication, in: 2021 1st International Conference on Emerging Smart Technologies and Applications (eSmarTA), 2021, pp. 1–4. doi:10.1109/eSmarTA52612.2021.9515739.
- [27] K. T. Swami, A. A. Moghe, A Review of LiFi Technology, in: 2020 5th IEEE International Conference on Recent Advances and Innovations in Engineering (ICRAIE), 2020, pp. 1–5. doi:10.1109/ICRAIE51050. 2020.9358340.
- [28] K. L. Bober, V. Jungnickel, M. Emmelmann, M. Riegel, E. Tangdiongga, A. M. J. Koonen, J.-P. Linnartz, D. Behnke, M. Müller, P.-B. Bök, S. V. Colonques, M. M. Vazquez, A Flexible System Concept for LiFi in the Internet of Things, in: IEEE 87th Vehicular Technology Conference (VTC Spring), 2020, pp. 1–4. doi:10.1109/ICT0N51198.2020.9203185.
- [29] J. Patel, P. Trivedi, D. Patel, A Performance Analysis of "Light Fidelity" and "Internet of Things" and It's Application, in: 2017 International Conference on Transforming Engineering Education (ICTEE), 2017, pp. 1–4. doi:10.1109/ICTEED.2017.8585696.
- [30] P. Lorenz, L. Hamada, LiFi Towards 5G: Concepts Challenges Applications in Telemedecine, in: 2020 Second International Conference on Embedded and Distributed Systems (EDiS), 2020, pp. 123–127. doi: 10.1109/EDiS49545.2020.9296435.
- [31] M. Mekala, P. Viswanathan, N. Srinivasu, G. Varma, Accurate Decisionmaking System for Mining Environment using Li-Fi 5G Technology over IoT Framework, in: 2019 International Conference on contemporary Computing and Informatics (IC3I), 2019, pp. 74–79. doi:10.1109/ IC3I46837.2019.9055546.
- [32] J. Dinesh Kumar, K. Priyadharsini, K. Srinithi, R. Samprtiha, C. Ganesh Babu, An Experimental Analysis of Lifi and Deployment on Localization Based Services and Smart Building, in: 2021 International Conference on Emerging Smart Computing and Informatics (ESCI), 2021, pp. 92–97. doi:10.1109/ESCI50559.2021.9396889.
- [33] C. SaiSankeerth Reddy, G. SakethRam, C. SaiSumanth Reddy, B. Yasaswi, P. Pandey, Appliance Automation Using LiFi with MQTT in Hybrid Star and Tree Network Setup, in: 2018 International Conference on Research in Intelligent and Computing in Engineering (RICE), 2018, pp. 1–6. doi:10.1109/RICE.2018.8509085.
- [34] I. Demirkol, D. Camps-Mur, J. Paradells, M. Combalia, W. Popoola, H. Haas, Powering the Internet of Things through Light Communication, IEEE Communications Magazine 57 (6) (2019) 107–113. doi: 10.1109/MC0M.2019.1800429.
- [35] D. Manimaran, M. Anu, C. Steffi, F, Integration of Indoor and Outdoor Voice Based Navigation Detection using Light based Communication (Lifi) and IoT, in: 2021 5th International Conference on Computing Methodologies and Communication (ICCMC), 2021, pp. 400–403. doi:10.1109/ICCMC51019.2021.9418379.
- [36] R.A.A.Othman, D. ap Sagaran, M. Mokayef, W.I.I.R.B.W.M.Nasir, Effective LiFi communication for IoT applications, in: 2018 IEEE 4th International Symposium in Robotics and Manufacturing Automation

(ROMA), 2018, pp. 1-4. doi:10.1109/ROMA46407.2018.8986698.

- [37] Z. Zhou, M. Kavehrad, W. Wang, J. Jiang, Research on Optical Internetof-Things for Informalization Development of Modern Power Grids, in: 2020 IEEE/IAS Industrial and Commercial Power System Asia (I&CPS Asia), 2020, pp. 1480–1485. doi:10.1109/ICPSAsia48933.2020. 9208568.
- [38] P. K. Sharma, Y.-S. Jeong, J. H. Park, EH-HL: Effective Communication Model by Integrated EH-WSN and Hybrid LiFi/WiFi for IoT, IEEE Internet of Things Journal 5 (3) (2018) 1719–1726. doi:10.1109/JIOT. 2018.2791999.
- [39] I. Sulayman, R. He, M. Manka, A. Ning, A. Ouda, LiFi/WiFi Authentication and Handover Protocols: Survey, Evaluation, and Recommendation, in: 2021 International Symposium on Networks, Computers and Communications (ISNCC), 2021, pp. 1–6. doi:10.1109/ISNCC52172.2021. 9615853.
- [40] M. Asad, S. Qaisar, A. Basit, Client Based Access Layer QoS Provisioning in Beyond 5G IoT Networks, in: 2020 3rd International Conference on Advanced Communication Technologies and Networking (Comm-Net), 2020, pp. 1–8. doi:10.1109/CommNet49926.2020.9199612.
- [41] M. Asad, S. Qaisar, A. Basit, Client-Centric Access Device Selection for Heterogeneous QoS Requirements in Beyond 5G IoT Networks, IEEE Access 8 (2020) 219820–219836. doi:10.1109/ACCESS.2020. 3042522.
- [42] S. Alfattani, Review of LiFi Technology and Its Future Applications, Journal of Optical Communications 42 (1) (2021) 121–132. doi:doi: 10.1515/joc-2018-0025.
- [43] L. Pescosolido, E. Ancillotti, A. Passarella, Performance Evaluation of Switching Between WiFi and LiFi under a Common Virtual Network Interface, in: 2022 IEEE International Conference on Smart Computing (SMARTCOMP), 2022, pp. 275–280. doi:10.1109/ SMARTCOMP55677.2022.00067.
- [44] M. Asad, S. Qaisar, Energy Efficient QoS-Based Access Point Selection in Hybrid WiFi and LiFi IoT Networks, IEEE Transactions on Green Communications and Networking 6 (2) (2022) 897–906. doi: 10.1109/TGCN.2021.3115729.
- [45] H. Haas, LiFi is a paradigm-shifting 5G technology, Reviews in Physics 3 (2018) 26-31. doi:https://doi.org/10.1016/j.revip.2017. 10.001.
- [46] C. Zeyu, 6G, LIFI and WIFI Wireless Systems: Challenges, Development and Prospects, in: 2021 18th International Computer Conference on Wavelet Active Media Technology and Information Processing (IC-CWAMTIP), 2021, pp. 322–325. doi:10.1109/ICCWAMTIP53232. 2021.9674090.
- [47] X. Wu, D. C. O'Brien, Parallel Transmission LiFi, IEEE Transactions on Wireless Communications 19 (10) (2020) 6268–6276. doi:10.1109/ TWC.2020.3001983.
- [48] H. Nguyen, Y. M. Jang, Design of MIMO C-OOK using Matched filter for Optical Camera Communication System, in: 2021 International Conference on Artificial Intelligence in Information and Communication (ICAIIC), 2021, pp. 474–476. doi:10.1109/ICAIIC51459.2021. 9415204.
- [49] M. Arfaoui, M. Soltani, I. Tavakkolnia, A. Ghrayeb, M. Safari, C. Assi, H. Haas, Physical Layer Security for Visible Light Communication Systems: A Survey, IEEE Communications Surveys & Tutorials 22 (3) (2020) 1887–1908. doi:10.1109/C0MST.2020.2988615.
- [50] M. Kafafy, Y. Fahmy, M. Khairy, Multi-cell VLC system design under illumination and communication constraints, Optical Switching and Networking 34 (2019) 23–34. doi:https://doi.org/10.1016/j.osn. 2019.05.001.
- [51] A. B. Ozyurt, W. O. Popoola, LiFi-Based D2D Communication in Industrial IoT, IEEE Systems Journal (2022) 1–8doi:10.1109/JSYST. 2022.3179183.
- [52] N. Partheeban, V. Nagaraju, N. J. Kumar, A. M. Ali, B. R. T. Bapu, An Improved Internet of Things enabled Wireless Data Transmission using Visible Light Communication, in: 2022 International Conference on Advances in Computing, Communication and Applied Informatics (AC-CAI), 2022, pp. 1–7. doi:10.1109/ACCAI53970.2022.9752518.
- [53] H. Alshaer, H. Haas, O. Y. Kolawole, An Optimal Networked LiFi Access Point Slicing Scheme for Internet-of-Things, in: 2021 IEEE International Conference on Communications Workshops (ICC Workshops), 2021, pp. 1–6. doi:10.1109/ICCWorkshops50388.2021.9473868.

- [54] A. Duquel, R. Stanica, H. Rivano, A. Desportes, Decoding methods in LED-to-smartphone bidirectional communication for the IoT, in: 2018 Global LIFI Congress (GLC), 2018, pp. 1–6. doi:10.23919/GLC. 2018.8319118.
- [55] J. D. Bokefode, S. A. Ubale, R. M. Gaikwad, Retrieving Real Time Data Through IOT Devices and Storing Securely on Cloud Using Li-Fi, in: 2018 3rd International Conference for Convergence in Technology (I2CT), 2018, pp. 1–5. doi:10.1109/I2CT.2018.8529393.
- [56] S. M. Kouhini, Z. Ma, C. Kottke, S. M. Mana, R. Freund, V. Jungnickel, LiFi based Positioning for Indoor Scenarios, in: 2021 17th International Symposium on Wireless Communication Systems (ISWCS), 2021, pp. 1–5. doi:10.1109/ISWCS49558.2021.9562207.
- [57] S. M. Kouhini, C. Kottke, Z. Ma, R. Freund, V. Jungnickel, M. Müller, D. Behnke, M. M. Vazquez, J.-P. M. G. Linnartz, LiFi Positioning for Industry 4.0, IEEE Journal of Selected Topics in Quantum Electronics 27 (6) (2021) 1–15. doi:10.1109/JSTQE.2021.3095364.
- [58] V. Jungnickel, M. Hinrichs, K. Bober, C. Kottke, A. Corici, M. Emmelmann, J. Rufo, P.-B. Bök, D. Behnke, M. Riege, X. Wu, R. Singh, D. O'Brien, S. Collins, F. Faulkner, M. Vazquez, M. Bech, F. Geilhardt, R.-P. Braun, X. Deng, E. Tangdiongga, A. Koonen, Enhance Lighting for the Internet of Things, in: 2019 Global LIFI Congress (GLC), 2019, pp. 1–6. doi:10.1109/GLC.2019.8864126.
- [59] J. Linnartz, C. Corrêa, T. Cunha, E. Tangdiongga, T. Koonen, X. Deng, M. Wendt, A. Abbo, P. Stobbelaar, M. Müller, D. Behnke, M. Vazquez, S. Colonques, M. Bech, T. Metin, M. Emmelmann, S. Kouhini, K. Bober, C. Kottke, V. Jungnickel, ELIoT: New Features in LiFi for Next-Generation IoT, in: 2021 Joint European Conference on Networks and Communications & 6G Summit (EuCNC/6G Summit), 2021, pp. 148– 153. doi:10.1109/EuCNC/6GSummit51104.2021.9482478.
- [60] M. Chowdhury, M. Hasan, M. Shahjalal, E. Shin, Y. Jang, Opportunities of Optical Spectrum for Future Wireless Communications, in: 2019 International Conference on Artificial Intelligence in Information and Communication (ICAIIC), 2019, pp. 004–007. doi:10.1109/ICAIIC. 2019.8668981.
- [61] M. R. Ghaderi, LiFi and Hybrid WiFi/LiFi indoor networking: From theory to practice, Optical Switching and Networking 47 (2023) 100699. doi:https://doi.org/10.1016/j.osn.2022.100699.
- [62] X. Wu, M. D. Soltani, L. Zhou, M. Safari, H. Haas, Hybrid LiFi and WiFi Networks: A Survey, IEEE Communications Surveys & Tutorials 23 (2) (2021) 1398–1420. doi:10.1109/COMST.2021.3058296.
- [63] M. Popadić, E. Kočan, LiFi Networks: Concept, Standardization Activities and Perspectives, in: 2021 25th International Conference on Information Technology (IT), 2021, pp. 1–4. doi:10.1109/IT51528.2021. 9390098.
- [64] S. E. Nordholm, H.-J. Zepernick, Modulation and Detection, ELECTRI-CAL ENGINEERING - Vol. I (2005).
- [65] A. F. Hussein, H. Elgala, T. D. Little, Evolution of Multi-Tier Transmission Towards 5G Li-Fi Networks, in: 2018 IEEE Global Communications Conference (GLOBECOM), 2018, pp. 1–7. doi:10.1109/ GL0C0M.2018.8647846.
- [66] S. D. Dissanayake, J. Armstrong, Comparison of ACO-OFDM, DCO-OFDM and ADO-OFDM in IM/DD Systems, Journal of Lightwave Technology 31 (7) (2013) 1063–1072. doi:10.1109/JLT.2013.2241731.
- [67] J. Armstrong, B. J. Schmidt, Comparison of Asymmetrically Clipped Optical OFDM and DC-Biased Optical OFDM in AWGN, IEEE Communications Letters 12 (5) (2008) 343–345. doi:10.1109/LCOMM.2008. 080193.
- [68] R. Iwasaki, K. Ohuchi, PAPR Reduction in OFDM Signal by Combining Partial Transmit Sequences with Precoding Matrix, in: 2018 12th International Conference on Signal Processing and Communication Systems (ICSPCS), 2018, pp. 1–6. doi:10.1109/ICSPCS.2018.8631723.
- [69] M. Hinrichs, B. Poddig, M. Nölle, P. Hellwig, R. Freund, V. Jungnickel, Efficient Line Coding for Low-Power Optical Wireless Communications, in: 2021 IEEE 94th Vehicular Technology Conference (VTC2021-Fall), 2021, pp. 1–7. doi:10.1109/VTC2021-Fall52928.2021.9625557.
- [70] R. Raj, A. Dixit, An Energy-Efficient Power Allocation Scheme for NOMA-Based IoT Sensor Networks in 6G, IEEE Sensors Journal 22 (7) (2022) 7371–7384. doi:10.1109/JSEN.2022.3153314.
- [71] Z. Ding, X. Lei, G. Karagiannidis, R. Schober, J. Yuan, V. Bhargava, A Survey on Non-Orthogonal Multiple Access for 5G Networks: Research Challenges and Future Trends, IEEE Journal on Selected Areas

in Communications 35 (10) (2017) 2181–2195. doi:10.1109/JSAC. 2017.2725519.

- [72] L. Chettri, R. Bera, A Comprehensive Survey on Internet of Things (IoT) Toward 5G Wireless Systems, IEEE Internet of Things Journal 7 (1) (2020) 16–32. doi:10.1109/JIOT.2019.2948888.
- [73] R. Feng, C.-X. Wang, J. Huang, X. Gao, S. Salous, H. Haas, Classification and Comparison of Massive MIMO Propagation Channel Models, IEEE Internet of Things Journal 9 (23) (2022) 23452–23471. doi:10.1109/ JIOT.2022.3198690.
- [74] E. Larsson, O. Edfors, F. Tufvesson, T. Marzetta, Massive MIMO for next generation wireless systems, IEEE Communications Magazine 52 (2) (2014) 186–195. doi:10.1109/MC0M.2014.6736761.
- [75] S. Gunnarsson, J. Flordelis, L. V. D. Perre, F. Tufvesson, Channel Hardening in Massive MIMO: Model Parameters and Experimental Assessment, IEEE Open Journal of the Communications Society 1 (2020) 501–512. doi:10.1109/0JCOMS.2020.2987704.
- [76] S. M. Kouhini, J. Hohmann, S. Mavanchery Mana, P. Hellwig, D. Schulz, A. Paraskevopoulos, R. Freund, V. Jungnickel, All-Optical Distributed MIMO for LiFi: Spatial Diversity Versus Spatial Multiplexing, IEEE Access 10 (2022) 102646–102658. doi:10.1109/ACCESS.2022. 3207475.
- [77] M. Uysal, F. Miramirkhani, T. Baykas, N. Serafimovski, V. Jungnickel, IEEE 802.11bb Reference Channel Models for Indoor Environments, Tech. rep., IEEE, Istanbul, Turkey (July 2018).
- [78] A. A. Purwita, H. Haas, Studies of Flatness of LiFi Channel for IEEE 802.11bb, in: 2020 IEEE Wireless Communications and Networking Conference (WCNC), 2020, pp. 1–6. doi:10.1109/WCNC45663.2020. 9120811.
- [79] ITU, High-speed indoor visible light communication transceiver System architecture, physical layer and data link layer specification, Recommendation, ITU-T G.9991 (Mar. 2019).
- [80] H. B. Eldeeb, S. M. Mana, V. Jungnickel, P. Hellwig, J. Hilt, M. Uysal, Distributed MIMO for Li-Fi: Channel Measurements, Ray Tracing and Throughput Analysis, IEEE Photonics Technology Letters 33 (16) (2021) 916–919. doi:10.1109/LPT.2021.3072254.
- [81] S. M. Mana, K. G. K. Gabra, S. M. Kouhini, P. Hellwig, J. Hilt, V. Jungnickel, An Efficient Multi-Link Channel Model for LiFi, in: 2021 IEEE 32nd Annual International Symposium on Personal, Indoor and Mobile Radio Communications (PIMRC), 2021, pp. 1–6. doi:10.1109/ PIMRC50174.2021.9569661.
- [82] S. M. Mana, K. G. K. Gabra, S. M. Kouhini, M. Hinrichs, D. Schulz, R. Freund, V. Jungnickel, LIDAR-Assisted Channel Modelling for LiFi, in: 2022 Optical Fiber Communications Conference and Exhibition (OFC), 2022, pp. 1–3.
- [83] M. Dehghani Soltani, A. A. Purwita, I. Tavakkolnia, H. Haas, M. Safari, Impact of Device Orientation on Error Performance of LiFi Systems, IEEE Access 7 (2019) 41690–41701. doi:10.1109/ACCESS.2019. 2907463.
- [84] M. D. Soltani, Z. Zeng, I. Tavakkolnia, H. Haas, M. Safari, Random Receiver Orientation Effect on Channel Gain in LiFi Systems, in: 2019 IEEE Wireless Communications and Networking Conference (WCNC), 2019, pp. 1–6. doi:10.1109/WCNC.2019.8886097.
- [85] X. Wang, C. Gan, S. Wu, Y. Chen, Y. Chen, Founding AP's IVM and related IHM for subway LiFi network, Optical Switching and Networking 45 (2022) 100676. doi:https://doi.org/10.1016/j.osn.2022. 100676.